\begin{table}%t4
\caption{Fundamental parameters of selected open clusters.}\label{parameters}
%\centerline
{
\begin{tabular}{lrrrccrcrrrr}
\hline\hline
Star cluster & {\it l} \hspace{0.25cm} & b \hspace{0.15cm} & $d$ 
\hspace{0.15cm} & $E(B-V)$ & Age & [Fe/H] & References & $X$\hspace{0.15cm} & 
$Y$\hspace{0.15cm} & $Z$\hspace{0.15cm} & $R_{\rm GC}$ \\
& ($\degr$)\hspace{0.15cm} & ($\degr$)\hspace{0.15cm} & (kpc) & (mag) & (Gyr) 
&  & & (kpc) & (kpc) & (kpc) & (kpc) \\
\hline
NGC~1342     & 154.99 & $-$15.38 & 0.63 & 0.30 & 0.50 & $-$0.24 & 1, 2   &  9.05 &
0.26 & $-$0.17 & 9.08 \\
NGC~1758     & 179.15 & $-$10.52 & 0.76 & 0.34 & 0.40 &  0.00 & 5     &  9.25 &
0.01 & $-$0.14 & 9.26 \\
NGC~1907     & 172.62 & +00.30 & 1.78 & 0.40 & 0.40 & $-$0.20 & 12, 13 & 10.27 &
0.23 & 0.01 & 10.27 \\
NGC~2099     & 177.65 & +03.09 & 1.40 & 0.30 & 0.40 &  0.09 & 4, 11  &  9.90 &
0.06 & 0.07   & 9.90 \\
NGC~2194     & 197.26 & $-$02.33 & 3.20 & 0.55 & 0.40 & $-$0.27 & 10    & 11.55 &
$-$0.95& $-$0.13   & 11.59 \\
NGC~2324     & 213.44 & +03.32 & 3.80 & 0.25 & 0.44 & $-$0.31 & 14    & 11.66 &
$-$2.09& 0.22   & 11.86 \\
NGC~2447     & 240.07 & +00.15 & 1.06 & 0.04 & 0.45 &  0.03 & 3     &  9.03 &
$-$0.92& 0.00   & 9.08 \\
NGC~2482     & 241.63 & +01.97 & 1.00 & 0.10 & 0.40 &  0.13 & 2, 4   &  8.98 & 
$-$0.88& 0.03   & 9.02 \\
NGC~5999     & 326.02 & $-$01.93 & 2.20 & 0.45 & 0.40 &  0.18 & 6, 7   &  6.68 &
$-$1.23& $-$0.07   & 6.79 \\
NGC~6631     &  19.47 & $-$00.19 & 2.60 & 0.60 & 0.40 &  0.05 & 8     &  6.05 &
0.87 & $-$0.01   & 6.11 \\
Melotte~111  & 221.20 & +84.01 & 0.09 & 0.00 & 0.50 & $-$0.06 & 9     &  8.51 &
$-$0.01 & 0.09  & 8.57 \\
\hline
\end{tabular}}
\medskip
{\sc References:} (1) \citet{jh75}; (2) \citet{pca95};
(3) \citet{hetal00}; (4) \citet{taa97}; (5) 
\citet{geetal98}; (6) \citet{petal99}; (7) \citet{sb93}; (8) \citet{setal01}; (9) 
\citet{oetal98}; (10) Paper~I; (11) \citet{ns02}; (12) \citet{s91}; (13) \citet{ss99}; (14) this study.
\end{table}